Author's response to reviews

Title: A cluster randomised controlled trial of the clinical and cost-effectiveness of a 'whole systems' model of self-management support for the management of long-term conditions in primary care: trial protocol

Authors:

Peter Bower (peter.bower@manchester.ac.uk)
Anne Kennedy (anne.p.kennedy@manchester.ac.uk)
David Reeves (david.reeves@manchester.ac.uk)
Anne Rogers (anne.rogers@manchester.ac.uk)
Tom Blakeman (tom.blakeman@manchester.ac.uk)
Carolyn Chew-Graham (carolyn.chew-graham@manchester.ac.uk)
Robert Bowen (robert.bowen@manchester.ac.uk)
Martin Eden (martin.eden@manchester.ac.uk)
Caroline Gardner (caroline.j.gardner@manchester.ac.uk)
Mark Hann (mark.hann@manchester.ac.uk)
Victoria Lee (victoria.lee@manchester.ac.uk)
Rebecca Morris (rebecca.morris@manchester.ac.uk)
Joanne Protheroe (joanne.protheroe@manchester.ac.uk)
Gerry Richardson (gar2@york.ac.uk)
Caroline Sanders (caroline.sanders@manchester.ac.uk)
Angela Swallow (angela.swallow@manchester.ac.uk)
David Thompson (david.thompson@manchester.ac.uk)

Version: 7 Date: 23 January 2012

Author's response to reviews:

The 3 tables have now been removed and are now presented as additional files in the text and at the end of the manuscript as suggested by the production editor.